

Exhumation History Along the Muli Thrust-Implication for Crustal Thickening Mechanism in Eastern Tibet

Paul Pitard, Anne Replumaz, Marie-luce Chevalier, Philippe-Hervé Leloup, Mingkun Bai, Marie-Pierre Doin, Cédric Thieulot, Xiong Ou, Mélanie Balvay, Haibing Li

► To cite this version:

Paul Pitard, Anne Replumaz, Marie-luce Chevalier, Philippe-Hervé Leloup, Mingkun Bai, et al.. Exhumation History Along the Muli Thrust-Implication for Crustal Thickening Mechanism in Eastern Tibet. Geophysical Research Letters, 2021, 48 (14), pp.e2021GL093677. 10.1029/2021gl093677. hal-03371286v2

HAL Id: hal-03371286 https://univ-lyon1.hal.science/hal-03371286v2

Submitted on 8 Oct 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

All Rights Reserved.

© 2021, American Geophysical Union.

Geophysical Research Letters

RESEARCH LETTER

10.1029/2021GL093677

Key Points:

- Thermo-kinematic modeling of Muli thrust, a major thrust fault of SE Tibetan Plateau
- 15 km crust exhumation in 50 Ma on a high-angle (>70°) ramp—décollement fault linked to thickening of SE Tibetan crust
- Fault related to significant Moho step and shear wave velocity contrast in deep crust suggests entire crust implication

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

P. Pitard, paul.pitard@univ-grenoble-alpes.fr

Citation:

Pitard, P., Replumaz, A., Chevalier, M.-L., Leloup, P.-H., Bai, M., Doin, M.-P., et al. (2021). Exhumation history along the Muli thrust—Implication for crustal thickening mechanism in eastern Tibet. *Geophysical Research Letters*, 48, e2021GL093677. https://doi. org/10.1029/2021GL093677

Received 2 APR 2021 Accepted 24 JUN 2021 Exhumation History Along the Muli Thrust—Implication for Crustal Thickening Mechanism in Eastern Tibet

P. Pitard¹, A. Replumaz¹, M.-L. Chevalier^{2,3}, P.-H. Leloup⁴, M. Bai², M.-P. Doin¹, C. Thieulot⁵, X. Ou¹, M. Balvay¹, and H. Li^{2,3}

¹CNRS, USMB, IRD, IFSTTAR, ISTerre, Université Grenoble Alpes, Grenoble, France, ²Institute of Geology, Chinese Academy of Geological Sciences, Beijing, China, ³Southern Marine Science and Engineering Guangdong Laboratory (Guangzhou), Guangzhou, China, ⁴Laboratoire de Géologie de Lyon, CNRS UMR 5570, Université de Lyon 1, Villeurbanne, France, ⁵Mantle Dynamics and Theoretical Geophysics, Utrecht University, Utrecht, The Netherlands

Abstract Thrusting implication in the crustal thickening history of eastern Tibet is highly debated. The ~250 km-long Muli thrust of the Yalong thrust belt in SE Tibet is a major Miocene structure with a pronounced topographic step (~2,000 m). Using thermo-kinematic modeling based on thermochronology data, we constrain the crustal geometry of the thrust as being steep (>70°) at the surface, in agreement with field observations, and flattening at depth (\geq 20 km) on an intra-crustal décollement. Thrusting motion on the fault shows a velocity of 0.2 ± 0.06 km/Ma since 50 Ma, followed by an acceleration at a rate of 0.6 ± 0.08 km/Ma starting at 12.5 ± 1 Ma, yielding a total of ~15 km of exhumed crust. Deeper, deformation may be localized through a ductile shear zone, and be related to the ~15 km Moho step and shear wave velocity contrast imaged by tomography beneath the Yalong thrust belt.

Plain Language Summary The India-Eurasia collision (~50 million years ago [Ma]) led to the formation of the Tibetan Plateau, the world's largest and highest orogenic plateau. The formation and evolution of such a unique geological feature has been one of the main controversies in Earth Sciences for decades, especially regarding the role of faulting in the thickening of the crust. Here, we present 3D thermo-kinematic models of thermochronology data allowing to constrain the exhumation history of the Muli thrust fault, a ~250 km-long major structure of the SE Tibetan margin, linked to significant steps in surface topography and in crustal boundary at depth (Moho). We constrain a steep fault (>70°) within the upper crust, consistent with field observations, that flattens at depth (\geq 20 km). The Muli thrust presents rapid thrusting motion (0.6 ± 0.08 km/Ma) that initiated at ~12.5 Ma, following a slower phase (0.2 ± 0.06 km/Ma) since 50 Ma, with total rock exhumation of ~15 km. This underlines the important role of thrust faulting in the thickening of the SE Tibetan crust.

1. Introduction

The eastern margin of the world's largest and highest orogenic plateau, the Tibetan Plateau, shows contrasted edges, with a steep topographic gradient along the Longmen Shan (LMS) north of the Xianshuihe strike-slip fault (XF), and gentler slopes across the Yalong margin to the south (Figure 1a). The driving mechanism(s) leading to such contrasts between margins are still poorly understood and have given rise to end-member models of plateau growth with different crustal structures: either a weak lower crust localizing crustal thickening ("channel flow model," e.g., Clark & Royden, 2000), or a more rigid crust, with thickening that includes the upper crust, in a brittle manner ("stepwise model," Tapponnier et al., 2001). Those two end-member models can be tested by investigating the role of thrusting mechanisms in SE Tibet.

At present, upper crustal deformation of the LMS is dominated by reverse faulting with a small component of right-lateral motion, as shown by recent earthquake focal mechanisms such as that of the 2008 Mw7.9 Wenchuan earthquake (e.g., de Michele et al., 2010; Fielding et al., 2013). This event ruptured two subparallel reverse structures, the Guanxian and Beichuan thrusts (Figure 1b), with steep shallow ramps (up to 70°) which root between 15 and 25 km-depth on a sub-horizontal décollement zone (e.g., Airaghi et al., 2017; Fielding et al., 2013; Hubbard & Shaw, 2009; Li et al., 2013; Qi et al., 2011) corresponding to the brittle-ductile transition (Feng et al., 2016). In addition, receiver function profiles (Robert et al., 2010) and regional shear waves tomography images (Yao et al., 2008) show a ~20 km step of the Moho depth between the







Figure 1. (a) Topographic map of Tibet (YM: Yalong Margin, LMS: Longmen Shan, SB: Sichuan Basin). (b) Geologic map of SE Tibet (CIGMR, 2004) with Mw >5 earthquake focal mechanisms, including 2008 Wenchuan earthquake (Mw7.9) in orange (Global CMT catalog). YTB: Yalong thrust belt, SZ: shear zone, GS: Gongga Shan. (c) Swath elevation profile across YTB.

plateau (\sim 60–63 km) and the Sichuan craton (\sim 40 km) at the apex of the Wenchuan shear zone. This has been interpreted either as a lower crustal ductile continuation of the shear zone (Robert et al., 2010), or as the halt of the lower crustal flow by the rigid Sichuan craton, hence inducing upward motion along the thrusts (Burchfiel et al., 2008).

By contrast, the Yalong margin only shows normal and strike-slip focal mechanisms, as well as two active left-lateral strike-slip fault systems: XF (Allen et al., 1991) and Litang (Chevalier et al., 2016; Wang & Burchfiel, 2000). Major geologic discontinuities attest, however, to the presence of the NE to EW-trending Yalong thrust belt (YTB) along the margin (e.g., Clark et al., 2005; Liu-Zeng et al., 2008; Perrineau, 2010; Wang & Burchfiel, 2000), consisting of the Muli and Jinhe-Qinghe thrusts that merge south of the XF (Figure 1b). Being most likely currently inactive considering regional seismicity, these thrusts were however active during the Miocene, as shown by thermochronology ages (Clark et al., 2005) as well as by the significant topographic step ($\sim 2,000$ m) (Figure 1c). The YTB has been identified as the SW continuation of the LMS (Yan & Lin, 2015) prior to activation of the XF at ~9 Ma (Y.-Z. Zhang et al., 2017), although the amount of thrusting remains controversial. Similarly to what has been observed in the LMS, tomography images (Yao et al., 2008) reveal a ~15 km vertical jump in crustal thickness across the YTB between the plateau (~60-63 km) and the Yalong margin (45-50 km) (Figure S5), coinciding with the limit of a low shear velocity layer in the middle and lower crusts (Figure S6), suggesting that the YTB plays a major role in thickening of the Yalong margin. By contrast, the channel flow model predicts a mainly horizontal flow at depth across the Yalong margin, generating a smooth topographic gradient with no spatial discontinuity and no thickening of the upper crust (Clark & Royden, 2000).

Whether the Yalong margin has been uplifted by major thrusts or has only been passively uplifted, is therefore key to understand the processes involved in the formation of SE Tibet and distinguish between these models. In this paper, we focus on the structure, amount and timing of exhumation of the Muli thrust, the northernmost segment of the YTB. We use thermo-kinematic modeling of low-temperature thermochronology data with apatite and zircon (U-Th)/He (AHe and ZHe respectively), and apatite fission-track (AFT) ages, as well as high-temperature systems with 40 Ar/ 39 Ar on biotite (BAr) and muscovite (MAr) from the hanging wall.

2. Geological Setting and Timing of Exhumation

The NW-dipping Muli thrust runs southward for ~250 km from the XF, with mainly Paleozoic resting atop Mesozoic rocks (Figure 1b). The thrust corresponds to a large topographic step bounding to the east the low-relief, high (>4,000 m) Tibetan Plateau, as well as a sharp bend of the Yalong River (Figure 1). There, the hanging wall consists of schists and marbles, mapped as Paleozoic and Triassic, with NNW to W-trending folds, with N70-130°-trending schistosity, intruded by numerous Cretaceous granites such as that of Datan (Figure 2a). Along the mapped trace of the thrust, schists and marbles have a strong schistosity and foliation trending N30° and steeply dipping 76°W on average, within a ~2 km-wide zone. Locally, the schistosity has a sigmoid geometry with slightly shallower (74°W) shear planes indicative of reverse shear (Figure 2b). In the footwall, sediments are mapped as Triassic and Upper Paleozoic and are only intruded by one small Mesozoic granite (Figure 1b).

Published AHe and AFT ages in the hanging wall of the Muli thrust have been interpreted as indicating fast exhumation between 9 and 13 Ma (Clark et al., 2005), while 80 km NW of the Muli thrust, AHe and ZHe





Figure 2. (a) Geologic map of Yalong River bend region (CIGMR, 2004), atop Landsat image, with sample locations. (b) Outcrop along Muli fault and stereographic plot (inset) showing sigmoidal schistosity (S1) between steep shear planes (c) indicating reverse shear. (c) Photo of deeply incised Yalong River.

ages have been interpreted to indicate the reactivation of the Jiulong thrust between 6 and 9 Ma, after an initial rapid exhumation phase between 35 and 30 Ma (H. Zhang et al., 2016). In the Gongga Shan massif, AFT and ZFT ages have been interpreted as representing the local activation of the XF at 9 Ma and its southward propagation along the Anninghe/Zemuhe fault at ~4 Ma (Y.-Z. Zhang et al., 2017).

3. Results

3.1. Thermochronology Data

Because of the steep banks of the Yalong River (Figure 2c), and the lack of granite outcrops in the footwall, the existing data set is limited to valley bottoms in the hanging wall of the Muli thrust. We here add three samples along a steep profile between 1,590 and 1,873 m of elevation within the Datan granite (Figure 2a), which yield AFT ages of 4.1 ± 0.9 , 5.0 ± 0.6 , and 8.8 ± 0.9 Ma (Figure 3, Table S1 and Text S1). To best constrain the crust's cooling history using thermochronology dating, one needs an adequate spatial distribution of the data and the use of several distinct thermochronology systems. We, therefore, compile all available ages (seven AHe, three AFT, three ZHe, two BAr, and two MAr) from previous studies (Clark et al., 2005; Perrineau, 2010; Wallis et al., 2003) (Table S2 and Text S2). Samples are ideally distributed along a 50 km-long transect perpendicular to the Muli thrust following the Jiulong River, at elevations between 1,557 and 2,624 m (Figure 2a). AHe (4.1 ± 0.2 to 13.9 ± 0.8 Ma) and ZHe ages (13.2 ± 0.8 to 37.2 ± 2.2 Ma) show a positive correlation with elevation and distance away from the fault (Figure 3). AFT ages are all located close to the junction between the Jiulong and Yalong Rivers (Figure 2a) and are tightly distributed between 4.7 ± 1.3 and 6.4 ± 1.1 Ma (Figure 3). BAr and MAr ages are Cretaceous, ranging from 104.3 ± 1.4 to 136 ± 7 Ma.





Figure 3. Comparison between measured and predicted ages on vertical (a, c, e, and g) and horizontal (b, d, f, and h) profiles perpendicular to the Muli fault for different scenarios.

3.2. Thermo-Kinematic Modeling

We use the code PECUBE (Braun, 2003; Braun et al., 2012) to unravel the 3D temporal evolution of the crust's thermal history. This code was designed to calculate theoretical ages for various thermochronometers, yielding the misfit value (Equation S1) with the measured ages for different scenarios, and to use an inversion method to find the model with the lowest misfit value. We define a $\sim 100 \times 100 \times 50$ km block with present-day surface topography and set a basal temperature of 800°C, corresponding to a geothermal gradient of $\sim 35^{\circ}$ C/km (Clark et al., 2005; Y.-Z. Zhang et al., 2015) (Tables S3, S4, and Text S3).

3.3. Crustal Thermal History Driven by Vertical Exhumation

We first conduct an inversion with more than 3,000 models (Table S3) to find the best fitting model that constrains the exhumation history for a scenario with a homogeneous vertical rock uplift. Our model starts at 140 Ma, slightly earlier than our oldest 40 Ar/ 39 Ar age. The best fitting exhumation scenario (misfit = 7.76) is a two-stages history with a first stage having a denudation rate of 0.033 km/Ma, followed at 14 Ma by a second stage with a faster rate of 0.47 km/Ma (Figure S1). This yields a total exhumation of 10.7 km since 140 Ma, including 6.5 km since 14 Ma. Predicted AHe ages are consistent with those measured for the low-elevation samples located <10 km away from the fault trace, while the four samples located at higher altitude, that is, farther away from the fault, show predicted AHe ages that are much younger (up to ~5 Ma) than the measured ones (Figures 3a and 3b). The difference is even higher for the three ZHe predicted ages, that is, up to ~15 Ma younger for samples farther from the fault. Those large differences suggest that a scenario with a homogeneous vertical uplift, and thus no fault, cannot fit the spatial distribution of our thermochronology data set (Figures 4a and 4b).

3.4. Crustal Thermal History Driven by the Muli Thrust

To simulate a more complex uplift velocity field, we implement a fault with exhumation of a hanging wall with respect to a stable footwall. The fault trace at the surface follows that of the Muli thrust (Figure 2a), and the number of segments defining the fault geometry at depth is set to three, which appears sufficient to get a satisfying velocity field. Given the lack of geophysical data constraining the deeper part of the fault, we run three distinct inversions in which we fix the deepest segment to be horizontal at depths of 20, 30, and 50 km. We add three variables to invert, in order to define the Muli thrust geometry, and only consider the low-temperature data (AHe, AFT, ZHe, ages <40 Ma), removing 40 Ar/ 39 Ar ages to reduce the number of thermochronometers used, and the time window of 50 Ma, which allows to reach a well-constrained inversion by running twice as many models (>6,000) than in our previous vertical exhumation setting.

The lowest misfit (2.02) inversion model is obtained for a steep (~80°) shallow fault segment down to 46 km-depth and a transitional segment from 46 to 50 km-depth dipping at ~10°, which joins the deep horizontal segment at 50 km-depth (Figures 4g and 4h). With this geometry, we find that the best fitting exhumation scenario occurred in two stages: a first stage with a fault velocity rate of 0.26 km/Ma, followed at 11.5 Ma by a second stage with a faster rate of 0.52 km/Ma (Figure S4). The modeled fault geometry yields an age zonation parallel to the fault, with young ages close to the fault, rapidly becoming older away from it (Figure 4). Young ages result from rapid vertical rock uplift above the steeply dipping fault plane, while away from the fault, older ages result from a slower rock uplift above the gently-dipping fault plane at depth. This thermo-kinematic model successfully reproduces the positive correlation between AHe and ZHe ages and distance to the fault (Figures 3g and 3h), with predicted AHe ages falling within uncertainty of the measured ones, and predicted ZHe ages evolving almost linearly with distance to the fault, reproducing well the measured ages. This spatial distribution is not obtained in our vertical exhumation scenario, and highlights the mandatory role of a listric Muli thrust.

Nevertheless, while the first order geometry of the fault is well constrained, the depth of the deep, flatter segment (equivalent to a basal décollement) is not. Indeed, running a model with a basal décollement at 30 km results in a misfit value being only slightly higher (2.12) than in our previous model using a basal décollement of 50 km (2.02). In that case, the shallow fault segment is dipping at 68°, the transitional segment is sub-horizontal (dip of 3°) (Figures 4e and 4f), and the transition time occurs 2 Ma earlier (13.5 Ma), separating two exhumation phases with velocity rates on the fault nearly identical to those in our previous



Geophysical Research Letters



Figure 4. Crustal thermal models for different scenarios overlaid by predicted present-day AHe ages wrapped on topography, with circles showing measured ages. Squares represent inflexion points defining fault geometry (yellow: inverted points; orange: fixed points): (a, c, e, and g) 3D view, (b, d, f, and h) 2D rock particle velocities (arrows) along middle section during most recent exhumation phase.

model (0.24 and 0.54 km/Ma) (Figure S3). The predicted AHe ages in such a scenario are slightly younger than those predicted for the 50-km basal depth model, while the ZHe ages show a more pronounced break in slope with distance to the fault, also consistent with the measured ages (Figures 3e and 3f). The misfit value (3.06) significantly increases for a model with a basal décollement depth of 20 km. In that case, the shallow fault segment is sub-vertical (~85°) down to 9 km-depth, the transitional segment is less steep (25°) from 9 to 20 km-depth (Figures 4c and 4d), and the transition time is 12.7 Ma, with a more pronounced rate jump from 0.14 to 0.67 km/Ma (Figure S2). The predicted AHe ages are ~2 Ma younger than the measured ones, and the predicted ZHe ages are up to 4 Ma older than the measured ages, leading to a higher misfit than in the previous models (Figures 3c and 3d).

4. Implication for the Crustal Thickening History of Eastern Tibet

Using both vertical and horizontal thermochronology data profiles in the hanging wall of the Muli thrust, we show that a scenario with a spatially homogeneous vertical uplift is incompatible with the currently available thermochronology data set. A more complex velocity field driven by a high-angle ramp-flat décollement fault geometry is required (Figure 4). Using data inversion, we obtain a ramp that dips between 85° and 68°, in agreement with surface field observations (Figure 2b), as well as with the high-angle (70°) ramp imaged in the LMS, for décollement depths of 20, 30 and 50 km that we test in agreement with LMS observations (Feng et al., 2016; Fielding et al., 2013; Li et al., 2013; Qi et al., 2011; Robert et al., 2010). The upper crustal velocity field in all three models show similar general kinematics with nearly vertical motion close to the fault leading to young predicted ages, and a more oblique, slower velocity farther away from the fault vielding older predicted ages, consistent with the measured ages (Figure 3). The exhumation evolution is also very similar between the three models, with a slow velocity rate on the fault between ~ 0.14 and ~ 0.25 km/Ma from 50 to 12.5 ± 1 Ma, followed by a rapid phase between ~ 0.53 and ~ 0.67 km/Ma. The early exhumation phase may be related to that recorded in the hanging wall of the Jiulong thrust between 35 and 30 Ma (H. Zhang et al., 2016), and to that in the LMS between 30 and 20 Ma (Tan et al., 2014; Wang et al., 2012). In the LMS, a second phase of rapid exhumation started at \sim 11 Ma at a rate of \sim 0.65 km/Ma (Godard et al., 2009), similar to the fast exhumation we document here along the Muli thrust. This would confirm a continuity between the YTB and LMS prior to the activation of the XF at \sim 9 Ma, having since then offset them by ~60 km (Yan & Lin, 2015; Y.-Z. Zhang et al., 2017) (Figure 1b).

For a thrust rooted at 20 km-depth, predicted ages are slightly diverging from the measured ages (younger predicted AHe ages, older ZHe ages, Figures 3c and 3d), leading to a slightly higher misfit (3.06) than for deeper décollements (2.02-2.12). Nevertheless, considering the simplicity of the kinematics modeling associated to fault displacement rates, such higher misfit is insufficient to reject this model, and mostly shows that low-temperature thermochonology data are more adequate to constrain the upper crust thermal evolution than that of the lower crust. To further discuss the structure of the Muli thrust in the lower crust, we then compare our results with geophysical data and crustal structures proposed for the LMS. The 20-km depth model agrees with the depth of the brittle-ductile transition (Feng et al., 2016), considering that thickening during the Cenozoic collision is limited to the brittle upper crust (Airaghi et al., 2017). The 50 km-depth model agrees with the thickening of the entire crust in the hanging wall of the Muli thrust fault (Figure 4h), as proposed for the Wenchuan shear zone in the LMS (Robert et al., 2010). This scenario is the only one compatible with an abrupt step of the Moho depth observed beneath the Yalong margin (Yao et al., 2008) (Figure 5). Indeed, for the Yalong margin, such step cannot be explained by a rheological contrast, as invoked in the LMS between the rigid Sichuan craton and the weak Tibetan crust, inducing a strong thickening of the Tibetan lower crust halted by the craton (e.g., Burchfiel et al., 2008). Beneath the Yalong margin, only a listric geometry of the Muli thrust rooting deep into the lower crust may induce a Moho step, the eastward limit of a low velocity crustal layer (Figure 5), and the thickening of the entire crust (Figure 4h). Below 20 km-depth, deformation has to be more ductile and localized in a ductile shear zone aligned with the upper crustal Muli thrust, which, since 50 Ma, exhumed a total of ~15 km of crust along its high-angle ramp and ~8 km above its sub-horizontal deep décollement, forming a significant topographic step of ~ 2 km (Figures 1c and 5).





Figure 5. 3D block diagram of Muli crustal fault geometry showing surface topography (2X vertical exaggeration), 20 km-depth brittle-ductile transition ("~" symbol), and tomography section at 50 km-depth (Yao et al., 2008).

5. Conclusions

We present 3D crustal models of different cooling scenarios using a wide spatial distribution of thermochronology ages from the Muli fault's hanging wall. We show that such a data set can only be related to a complex velocity field generated along a high-angle ($>70^\circ$) ramp thrust fault at the tip of a horizontal décollement. In such a scenario, we obtain a total exhumation of ~15 km since ~50 Ma, including ~7.5 km since 12.5 Ma. Our results show that the upper crust is strongly involved in the thickening of the Tibetan crust in this part of Tibet, and that it is not passively uplifted as assumed in channel flow models. Even if the rooting depth of the Muli fault remains difficult to constrain using such thermal modeling, tomography images reveal a Moho step and a shear wave velocity contrast in the deep lower crust at the apex of the YTB, which cannot be related to rheological contrast as proposed beneath the LMS. We, therefore, interpret such discontinuities as evidence of the listric nature of the Muli thrust rooting deep into the lower crust, behaving as a ductile shear zone below 20 km-depth, allowing the thickening of the whole crust.

Data Availability Statement

Supplementary data reported in this study are given in the Supporting Information and all apatite fission tracks data are stored in Figshare repository: https://doi.org/10.6084/m9.figshare.14770752. Supplementary thermochronology data used in this study can be found in Clark et al. (2005) (https://doi.org/10.1130/G21265.1) and Perrineau (2010) (http://www.theses.fr/2010GLOB0007).

References

Airaghi, L., de Sigoyer, J., Lanari, P., Guillot, S., Vidal, O., Monié, P., et al. (2017). Total exhumation across the Beichuan fault in the Longmen Shan (eastern Tibetan plateau, China): Constraints from petrology and thermobarometry. *Journal of Asian Earth Sciences*, 140, 108–121. https://doi.org/10.1016/j.jseaes.2017.04.003

Allen, C. R., Zhuoli, L., Hong, Q., Xueze, W., Huawei, Z., & Weishi, H. (1991). Field study of a highly active fault zone: The Xianshuihe fault of southwestern China. *Geological Society of America Bulletin*, 103, 1178–1199. https://doi.org/10.1130/0016-7606(1991)103%3C1 178:FSOAHA%3E2.3.CO;2

Braun, J. (2003). Pecube: A new finite-element code to solve the 3D heat transport equation including the effects of a time-varying, finite amplitude surface topography. *Computers & Geosciences*, 29(6), 787–794. https://doi.org/10.1016/S0098-3004(03)00052-9

Acknowledgments

This work was supported by the TelluS Program of CNRS/INSU, ISTerre, LGL-TPE, Agence Nationale de Recherche (ANR "Tibetan Orchestra"), National Natural Science Foundation of China [NSFC 42020104007, 41941016] and the Second Tibetan Plateau Scientific Expedition of the Ministry of Science and Technology of China [2019QZKK0901].



- Braun, J., van der Beek, P., Valla, P., Robert, X., Herman, F., Glotzbach, C., et al. (2012). Quantifying rates of landscape evolution and tectonic processes by thermochronology and numerical modeling of crustal heat transport using PECUBE. *Tectonophysics*, 524(525), 1–28. https://doi.org/10.1016/j.tecto.2011.12.035
- Burchfiel, B. C., Royden, L. H., van der Hilst, R. D., Hager, B. H., Chen, Z., King, R. W., et al. (2008). A geological and geophysical context for the Wenchuan earthquake of 12 May 2008, Sichuan, People's Republic of China. *Geological Society of America Today*, 18(7), 4. https:// doi.org/10.1130/GSATG18A.1
- Chengdu Institute of Geology and Mineral Resources (CIGMR). (2004). Geological map of Tibetan Plateau and its vicinity. Chengdu Map Press.
- Chevalier, M.-L., Leloup, P. H., Replumaz, A., Pan, J., Liu, D., Li, H., et al. (2016). Tectonic-geomorphology of the Litang fault system, SE Tibetan Plateau, and implication for regional seismic hazard. *Tectonophysics*, 682, 278–292. https://doi.org/10.1016/j.tecto.2016.05.039
- Clark, M. K., House, M. A., Royden, L. H., Whipple, K. X., Burchfiel, B. C., Zhang, X., & Tang, W. (2005). Late cenozoic uplift of southeastern Tibet. *Geology*, 33(6), 525. https://doi.org/10.1130/G21265.1
- Clark, M. K., & Royden, L. H. (2000). Topographic ooze: Building the eastern margin of Tibet by lower crustal flow. *Geology*, 28(8), 703–706. https://doi.org/10.1130/0091-7613(2000)28%3C703:TOBTEM%3E2.0.CO;2
- de Michele, M., Raucoules, D., de Sigoyer, J., Pubellier, M., & Chamot-Rooke, N. (2010). Three-dimensional surface displacement of the 2008 May 12 Sichuan earthquake (China) derived from Synthetic Aperture Radar: Evidence for rupture on a blind thrust: 3D surface displacement of Sichuan earthquake. *Geophysical Journal International*, 183(3), 1097–1103. https://doi.org/10.1111/j.1365-246X.2010.04807.x
- Feng, S., Zhang, P., Liu, B., Wang, M., Zhu, S., Ran, Y., et al. (2016). Deep crustal deformation of the Longmen Shan, eastern margin of the Tibetan Plateau, from seismic reflection and Finite Element modeling. *Journal of Geophysical Research: Solid Earth*, 121(2), 767–787. https://doi.org/10.1002/2015JB012352
- Fielding, E. J., Sladen, A., Li, Z., Avouac, J.-P., Bürgmann, R., & Ryder, I. (2013). Kinematic fault slip evolution source models of the 2008 M7.9 Wenchuan earthquake in China from SAR interferometry, GPS and teleseismic analysis and implications for Longmen Shan tectonics. *Geophysical Journal International*, 194(2), 1138–1166. https://doi.org/10.1093/gji/ggt155
- Godard, V., Pik, R., Lavé, J., Cattin, R., Tibari, B., de Sigoyer, J., et al. (2009). Late Cenozoic evolution of the central Longmen Shan, eastern Tibet: Insight from (U-Th)/He thermochronometry. *Tectonics*, 28(5), TC5009. https://doi.org/10.1029/2008TC002407
- Hubbard, J., & Shaw, J. H. (2009). Uplift of the Longmen Shan and Tibetan plateau, and the 2008 Wenchuan (*M* = 7.9) earthquake. *Nature*, 458(7235), 194–197. https://doi.org/10.1038/nature07837
- Li, H., Wang, H., Xu, Z., Si, J., Pei, J., Li, T., et al. (2013). Characteristics of the fault-related rocks, fault zones and the principal slip zone in the Wenchuan Earthquake Fault Scientific Drilling Project Hole-1 (WFSD-1). *Tectonophysics*, 584, 23–42. https://doi.org/10.1016/j. tecto.2012.08.021
- Liu-Zeng, J., Tapponnier, P., Gaudemer, Y., & Ding, L. (2008). Quantifying landscape differences across the Tibetan plateau: Implications for topographic relief evolution. *Journal of Geophysical Research*, 113(F4), F04018. https://doi.org/10.1029/2007JF000897

Perrineau, A. (2010). Evolution morphologique et tectonique récente des marges NE et SE du plateau tibétain: Lien avec la dynamique des grands fleuves (Doctoral dissertation). Institut de Physique du Globe. (in French).

- Qi, W., Xuejun, Q., Qigui, L., Freymueller, J., Shaomin, Y., Caijun, X., et al. (2011). Rupture of deep faults in the 2008 Wenchuan earthquake and uplift of the Longmen Shan. Nature Geoscience, 4(9), 634–640. https://doi.org/10.1038/ngeo1210
- Robert, A., Pubellier, M., de Sigoyer, J., Vergne, J., Lahfid, A., Cattin, R., et al. (2010). Structural and thermal characters of the Longmen Shan (Sichuan, China). *Tectonophysics*, 491(1–4), 165–173. https://doi.org/10.1016/j.tecto.2010.03.018
- Tan, X.-B., Lee, Y.-H., Chen, W.-Y., Cook, K. L., & Xu, X.-W. (2014). Exhumation history and faulting activity of the southern segment of the Longmen Shan, eastern Tibet. Journal of Asian Earth Sciences, 81, 91–104. https://doi.org/10.1016/j.jseaes.2013.12.002
- Tapponnier, P., Zhiqin, X., Roger, F., Meyer, B., Arnaud, N., Wittlinger, G., & Jingsui, Y. (2001). Oblique stepwise rise and growth of the Tibet Plateau. Science, 294(5547), 1671–1677. https://doi.org/10.1126/science.105978
- Wallis, S., Tsujimori, T., Aoya, M., Kawakami, T., Terada, K., Suzuki, K., & Hyodo, H. (2003). Cenozoic and Mesozoic metamorphism in the Longmenshan orogen: Implications for geodynamic models of eastern Tibet. *Geology*, 31(9), 745. https://doi.org/10.1130/G19562.1
- Wang, E., & Burchfiel, B. C. (2000). Late Cenozoic to Holocene deformation in southwestern Sichuan and adjacent Yunnan, China, and its role in formation of the southeastern part of the Tibetan Plateau. *Geological Society of America Bulletin*, 112(3), 413–423. https://doi.org/ 10.1130/0016-7606(2000)112%3C413:LCTHDI%3E2.0.CO;2
- Wang, E., Kirby, E., Furlong, K. P., van Soest, M., Xu, G., Shi, X., et al. (2012). Two-phase growth of high topography in eastern Tibet during the Cenozoic. *Nature Geoscience*, 5(9), 640–645. https://doi.org/10.1038/ngeo1538
- Yan, B., & Lin, A. (2015). Systematic deflection and offset of the Yangtze River drainage system along the strike-slip Ganzi-Yushu-Xianshuihe Fault Zone, Tibetan Plateau. Journal of Geodynamics, 87, 13–25. https://doi.org/10.1016/j.jog.2015.03.002
- Yao, H., Beghein, C., & van der Hilst, R. D. (2008). Surface wave array tomography in SE Tibet from ambient seismic noise and two-station analysis - II. Crustal and upper-mantle structure. *Geophysical Journal International*, 173(1), 205–219. https://doi. org/10.1111/j.1365-246X.2007.03696.x
- Zhang, H., Oskin, M. E., Liu-Zeng, J., Zhang, P., Reiners, P. W., & Xiao, P. (2016). Pulsed exhumation of interior eastern Tibet: Implications for relief generation mechanisms and the origin of high-elevation planation surfaces. *Earth and Planetary Science Letters*, 449, 176–185. https://doi.org/10.1016/j.epsl.2016.05.048
- Zhang, Y.-Z., Replumaz, A., Leloup, P. H., Wang, G.-C., Bernet, M., van der Beek, P., et al. (2017). Cooling history of the Gongga batholith: Implications for the Xianshuihe Fault and Miocene kinematics of SE Tibet. *Earth and Planetary Science Letters*, 465, 1–15. https://doi. org/10.1016/j.epsl.2017.02.025
- Zhang, Y.-Z., Replumaz, A., Wang, G.-C., Leloup, P. H., Gautheron, C., Bernet, M., et al. (2015). Timing and rate of exhumation along the Litang fault system, implication for fault reorganization in Southeast Tibet: Litang Fault System Exhumation History. *Tectonics*, 34(6), 1219–1243. https://doi.org/10.1002/2014TC003671

References From the Supporting Information

Clark, M. K. (2003). Late cenozoic uplift of southeastern Tibet (Doctoral dissertation). Massachusetts Institute of Technology.

Clark, M. K., Royden, L. H., Whipple, K. X., Burchfiel, B. C., Zhang, X., & Tang, W. (2006). Use of a regional, relict landscape to measure vertical deformation of the eastern Tibetan Plateau. *Journal of Geophysical Research: Earth Surface*, 111(F3), F03002. https://doi.org/10.1029/2005JF000294

- Ehlers, T. A., Chaudhri, T., Kumar, S., Fuller, C. W., Willett, S. D., Ketcham, R. A., et al. (2005). Computational tools for low-temperature thermochronometer interpretation. *Reviews in Mineralogy and Geochemistry*, 58(1), 589–622. https://doi.org/10.2138/rmg.2005.58.22 Hurford, A. J., & Green, P. F. (1983). The zeta age calibration of fission-track dating. *Chemical Geology*, 41, 285–317. https://doi.org/10.1016/
- S0009-2541(83)80026-6 Ouimet, W., Whipple, K., Royden, L., Reiners, P., Hodges, K., & Pringle, M. (2010). Regional incision of the eastern margin of the Tibetan Plateau. *Lithosphere*, 2(1), 50–63. https://doi.org/10.1130/L57.1
- Xu, G., & Kamp, P. J. J. (2000). Tectonics and denudation adjacent to the Xianshuihe Fault, eastern Tibetan Plateau: Constraints from fission track thermochronology. Journal of Geophysical Research: Solid Earth, 105(B8), 19231–19251. https://doi.org/10.1029/2000JB900159